

Low haemoglobin in arduous seasons is associated with reduced chance of ovulation among women living in the Bolivian altiplano

Rose Stevens, Alexandra Alvergne, Virginia J Vitzthum

► To cite this version:

Rose Stevens, Alexandra Alvergne, Virginia J Vitzthum. Low haemoglobin in arduous seasons is associated with reduced chance of ovulation among women living in the Bolivian altiplano. Evolution, Medicine, and Public Health, 2024, 12 (1), pp.191-203. 10.1093/emph/eoae022 . hal-04789964

HAL Id: hal-04789964 https://hal.umontpellier.fr/hal-04789964v1

Submitted on 19 Nov 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

Evolution, Medicine, and Public Health [2024] pp.191–203 https://doi.org/10.1093/emph/eoae022 Advance access date 12 September 2024

Low haemoglobin in arduous seasons is associated with reduced chance of ovulation among women living in the Bolivian *altiplano*



ORIGINAL

RESEARCH

MEDICINE, & PUBLIC HEALTH

Rose Stevens^{1,*,}, Alexandra Alvergne^{1,2} and Virginia J. Vitzthum^{3,4,}

¹School of Anthropology and Museum Ethnography, The University of Oxford, OXford, OX2 6PE, UK; ²ISEM, Univ Montpellier, CNRS, IRD, Montpellier, France; ³Department of Medicine, CEMCOR, University of British Columbia, Canada V5Z 1M9; ⁴Department of Anthropology, Indiana University, Bloomington, IN 47405, USA ^{*}Corresponding author: School of Anthropology and Museum Ethnography, 51/53 Banbury Road, The University of Oxford, Oxford OX2 6PE, UK. Tel: +44-78-09-22-01-16; E-mail: rose.stevens@anthro.ox.ac.uk Received 22 February 2024; revised version accepted 31 July 2024.

ABSTRACT

Background and objectives: Female reproductive function flexibly responds to ecological variation in energy availability, but the roles of other ecologically limited resources, such as iron, remain poorly understood. This analysis investigates whether haemoglobin associates with investment in reproductive function in a rural natural fertility population living in the Bolivian *altiplano*.

Methodology: We conducted a cross-sectional secondary analysis of prospectively collected biomarker and sociodemographic data, comprising 152 menstrual cycles from 96 non-contracepting women living at 3800 m altitude. Multivariable multilevel models were used to investigate (i) whether haemoglobin concentration is associated with ecological variation in subsistence strategy and seasonal conditions, and (ii) whether haemoglobin concentration is associated with the occurrence of ovulation and/or the concentration of luteal progesterone, two biomarkers of current investment in reproduction.

Results: Haemoglobin concentrations were lower in arduous seasons among those women more dependent on traditional agropastoral subsistence strategies ($\beta = -0.42$, 95% CI: -0.80 to -0.04, P = 0.032). During more arduous seasons, a 1 standard deviation increase in haemoglobin was associated with an over 3-fold increase in the odds of ovulation after adjusting for body fat, breastfeeding status, and age (adjusted odds ratio = 3.27, 95% CI: 1.10 to 9.27, P = 0.033).

© The Author(s) 2024. Published by Oxford University Press on behalf of the Foundation for Evolution, Medicine, and Public Health. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited. **Conclusions and implications**: When conditions are relatively harsh and may be expected to improve, low haemoglobin levels are associated with lower current investment in reproduction and reduced fecundity. These results support the role of iron, independent of energy stores, as a limiting resource in modulating reproductive trade-offs.

Lay summary Bolivian women with more traditional, agricultural lifestyles showed lower haemoglobin levels, a marker of iron status, during the most difficult seasonal conditions compared to better seasons. In the more difficult seasons, women who had lower haemoglobin levels were less likely to release an egg (ovulate) in that menstrual cycle.

KEYWORDS: haemoglobin; iron; anaemia; ovulation; seasonal ecology; life history trade-offs; longitudinal cohort study

1. INTRODUCTION

Iron is recognized as an essential nutrient, yet anaemia remains a major global health problem affecting nearly one-third of all non-pregnant women [1]. Maternal anaemia increases the risks of poor pregnancy outcomes including maternal and offspring death [2]. Various intervention efforts to reduce maternal anaemia have not been as successful as expected, often for unknown reasons [3]. Intervention strategies to meet the needs of women with maternal anaemia require a better understanding of the intricate connections between iron status and the underlying biological mechanisms governing fecundity. Here we use a reproductive ecology approach [4] and prospective physiological data from a traditional non-contracepting high fertility population to shed new light on the socio-ecological determinants of variation in iron status in women and the consequences of such variation for fecundity across socio-ecological contexts.

Iron, a limited resource [5] tightly regulated within the body (Box 1), is critical for both successful reproduction and maternal health. An offspring's survival requires that a mother transfers sufficient iron to the foetus during gestation, particularly as iron concentration in human milk is low. This substantial transfer of iron from mother to foetus throughout pregnancy depletes maternal iron stores [6], leaving women at elevated risk of anaemia during pregnancy. Maternal anaemia and low haemoglobin prior to and during pregnancy are associated with a host of negative outcomes for mother and baby [7], including preterm birth, poor maternal and foetal weight gain, low birth weight, growth faltering, and obstetric complications. For instance, Gambian



BOX 1: IRON IN THE BODY

Mechanistically, iron is a crucial resource for many biological functions, as many fundamental bodily processes evolved during a period when available iron was abundant in the environment [22]. Iron is predominantly used for synthesizing red blood cells, within which it plays a critical role in the binding of oxygen to haemoglobin for transportation around the body. Iron also has critical roles in enzymes used in respiration, as well as uses in cellular metabolism, survival, and proliferation. Iron is a particularly important bodily resource at high altitudes, where decreased oxygen availability requires physiological adjustments to allow for enough oxygen to reach the body's tissues [22]. In Andean populations, this adjustment is achieved through the mechanism of increased haemoglobin concentration among high-altitude residents [23] and as such cut offs for defining anaemia are higher in these populations [24].

Iron levels and storage must be carefully regulated as there is no active process that eliminates iron from the body and excess bioavailable iron can cause oxidative damage or can be used by pathogens for increased proliferation. Thus, most of the ~4–5 g of iron in the body is stored and its release is tightly regulated. Iron is typically stored as ferritin and transferrin, levels of which are controlled by hepcidin, a hormone that regulates absorption of iron from the gut and release of iron stores [23]. Thus, hepcidin controls how much iron is available as haemoglobin, which is the protein that carries oxygen in red blood cells and is the indicator used to measure anaemia status.

Low haemoglobin levels can be due to, amongst other things, (i) iron-deficiency anaemia, caused by nutritional deficiencies or blood loss, and (ii) anaemia of infection, caused by immune stress. While low haemoglobin indicates low bioavailability of iron, whether this represents low iron stores also depends on the type of anaemia. Iron deficiency anaemia is associated with low total iron stores (as identified through low haemoglobin and low ferritin). Anaemia of infection is associated with normal to high total iron stores (low haemoglobin and high ferritin), but in this case, iron is withheld from the blood, particularly when inflammation is high, to avoid utilization of iron by the infecting pathogens. Haemoglobin is one of the last indicators to drop in the case of iron deficiency, so low haemoglobin can either point to very low iron stores or the presence of an infection/high inflammation [7].

women with lower haemoglobin levels (defined as <12 g/dl) during the two years prior to giving birth had a greater risk of stillbirths and higher mortality rates amongst their live-born children [8]. Not only is iron associated with post-conception outcomes, it may be associated with risk of conception itself. In the same Gambian population, women with low haemoglobin levels prior to pregnancy also had longer interbirth-intervals [9]. In a Polish agricultural population, higher haemoglobin was associated with a thicker endometrium, which may increase the chance of successful implantation [10]. However, the relationship between fertility and iron may be U-shaped as women with the highest haemoglobin levels among the Tsimane in Bolivia showed a reduced chance of fertility across the study period [11]. There is little understanding of the mechanisms underlying the observed associations between iron and fecundity. Studies in non-clinical settings are rare, and the lack of a theoretical framework for understanding normal variation in reproductive outcomes in response to iron biology hinders our ability to inform interventions.

At the physiological level, reproductive ecology studies suggest that iron may regulate resource allocation trade-offs between current vs. future reproduction by influencing key processes such as energy metabolism, oxidative stress, and immune function [7]. Reproductive ecology research draws on life history theory [12, 13], a branch of evolutionary biology positing that organisms have limited resources for fulfilling the basic demands of reproduction, growth, and survival, leading to resource allocation trade-offs between essential fitness functions. While decades of human reproductive ecology research have shown that energy availability modulates reproductive trade-offs across a range of ecological settings [14-17], the importance of other limited resources, iron being a key candidate given its association with reproductive outcomes, remains under-researched [18]. That iron is involved in modulating the trade-off between current and future reproductive investment is suggested by a study in northern Kenya, where higher haemoglobin was associated with a greater chance of having resumed menstrual periods post-birth [19]. Iron has been shown to be necessary for the function of granulosa cells, which make up the follicle surrounding a maturing egg [20]. With reduced functioning of these cells, the ovaries will produce lower oestrogen levels [21], the chances of ovulation may decrease, and subsequent progesterone levels and endometrial growth would be lower. However, how iron modulates ovulation and levels of luteal progesterone, two markers of fecundity shaped by energy-dependent reproductive trade-offs across a range of ecological settings [4], is currently not well understood.

In this paper, we investigated the relationship between iron and markers of fecundity using prospective data on haemoglobin and progesterone levels from a unique natural fertility sample of Bolivian women. Our objectives are two-fold (i) to investigate if haemoglobin levels co-vary with women's socio-ecology, focussing on the influence of seasons and variation in wealth and subsistence strategies, and (ii) to investigate if variation in fecundity, specifically focussing on ovulation risk and luteal progesterone levels in a given menstrual cycle, is associated with haemoglobin levels. The occurrence of ovulation is a clear binary signal of fecundity because it reflects either minimal investment in, or irrevocable suppression of, reproduction in that cycle. Luteal progesterone levels, predominantly driven by secretions from the corpus luteum, drive endometrial decidualisation and successful implantation. Whilst there is little available research on the optimum absolute or relative endogenous levels of progesterone required for successful implantation, 'not enough' is considered a barrier to successful pregnancy and is referred to in medicine by the ambiguous term 'luteal phase deficiency (LPD)', though this is a contested 'disease' [22]. Our study sample overcomes many of the hurdles involved in human trade-off research because (i) there was no modern contraceptive use and all study participants were married or in unions, and (ii) iron is a particularly important and limited resource in this population in which haemoglobin levels are normally high as an adaption to high-altitude hypoxia [23-25]. These conditions may allow us to observe trade-offs that are masked among individuals living in circumstances with lower iron requirements.

2. METHODS

(a) Study population

The data for these analyses were originally collected as part of Project REPA (Reproduction and Ecology in Provincía Aroma), a longitudinal observational study among indigenous Aymara agropastoralists living in the Bolivian altiplano [26]. The data were collected from February to November 1996. These data were not collected to study the hypotheses tested here, therefore all data collectors and participants were blind to the present research questions. Women were from 30 rural communities scattered across approximately 400 km², situated at an average of 3800 m altitude, approximately half-way between La Paz and Oruro. Families in these communities were predominantly reliant on potatoes, barley, sheep, and/or dairy cattle. The only formal medical care available was a small clinic in the local town of Patacamaya, but this is a substantial distance from most women's homes. Inclusion criteria for Project REPA included falling within 20-40 years of age, currently in a stable heterosexual union, and not using contraception.

(b) Ethics

All study protocols, including consent procedures, were approved by the Consejo Scientifico, Instituto Boliviano de Biologia de Altura, La Paz, and the Institutional Review Board, University of California, Riverside. All study participants were asked for verbal informed consent, which was obtained in accordance with the cultural norms of the study population. Giving, or declining to give, consent was recorded in confidential field records.

(c) Analytical sample

The unit of analysis was a menstrual cycle, and cycles for which there were either hormone or anthropometric data were initially considered for analysis. This initial dataset comprised 817 menstrual cycles from 140 participants. As our focus was to investigate haemoglobin and fecundity, all cycles that were coded as pregnant or that did not have corresponding data for haemoglobin, ovulation, or progesterone were excluded from the analyses. Cycles with either a confirmed or possible early pregnancy loss (a positive hCG test followed by bleeding and a negative hCG test) were also excluded. Four cycles contained two measures of haemoglobin. For these cycles, the measurement taken latest in the cycle was retained in the dataset, as these cycles were often long and the latest measure was considered most likely to fall nearer to the putative ovulation date.

Miller's (2016) review on the ecology of iron in women suggests different predictions for changes in haemoglobin in response to hormonal changes across an individual menstrual cycle [7]. For instance, haemoglobin may decrease with increased progesterone (due to increased fluid retention), but also may increase with increased oestrogen due to oestrogen's blocking effect on hepcidin. An analysis of several variables of iron status across the menstrual cycle showed that whilst iron and hepcidin displayed a drop and rebound across menstruation followed by a stabilization in the second half of the cycle, haemoglobin levels did not consistently vary during the cycle [27]. We also found little evidence for variation in haemoglobin within menstrual cycles in our sample, and thus did not adjust for cycle day of the haemoglobin measurement. The final analytical sample consisted of 152 cycles from 96 participants (48 women contributed 1 cycle, 40 women contributed 2 cycles and 8 women contributed 3 cycles). The flowchart (Fig. 1) displays this process of sample creation.

(d) Treatment of missing values

Overall, around 3% of the data was missing, mostly for the economic strategy score. Analysis of complete cases only by dropping missing cases can introduce bias and lead to a substantial reduction of statistical power [28]. To avoid dropping these cases, imputation using random forest multiple imputation was conducted using the R package 'missRanger' [29], which combines random forest imputation with predictive mean matching. Prior to all analyses, we imputed 5 datasets, with a maximum of 10 iterations specified for each imputation. To provide the most information for making imputation predictions, imputation of missing data was calculated using a larger dataset (e.g., participants' weight, height, age at first birth) in addition to those variables that directly address our study questions. Parameter estimates for all five datasets were then pooled to provide more accurate estimates using the pool function from the R package 'mice' [30]. This function combines estimates from repeated complete data analyses into a single set of estimates and standard errors.

(e) Causal inference

Whilst causal inference is very difficult in observational studies, recently scholars have argued that there is merit in acknowledging our intention to investigate causality, making our assumptions



and limitations clear, and not shying away from the causal goal of research [31]. In order to think through which factors may be potential confounders and which to include as adjustment variables to help reduce bias in our analyses [32], we created a DAG (Directed Acyclic Graph) of hypothesized causal relationships [33] (Supplementary Figure 1). Based on this process, we minimally adjust for age for our first question, and age, body fat, and breastfeeding for our second question.

(f) Variables

Haemoglobin (g/dl) was measured using a HemoCue (Quest Diagnostics), a portable, battery-operated point-of-care testing device that requires only a single blood drop from a finger prick. This procedure was concurrent with the collection of anthropometrics, typically done once per cycle in each woman's home at a day and time arranged with her. Due to the presence of four low haemoglobin outliers, this variable was capped in order to meet the assumption of a normal distribution in our outcome variable in our multilevel linear models [34]. Capping takes any outliers below the fifth (11.8 g/dl) and above the 95th (17.5 g/dl) centiles and recodes their value to those of the 5th or 95th centiles, respectively. This allowed us to maintain potentially important observations close to their actual values and model them without losing quality in our models.

Ovulation is a binary variable (each cycle coded as ovulatory or anovulatory). Without ovulation there is no possibility of further reproductive investment during that cycle. Ovulation was ascribed to a given cycle using a previously published algorithm that is based on mean peak luteal progesterone concentration during the mid-luteal phase [26, 35].

Mean peak luteal progesterone (P4) concentration (pmol/l): Women were visited every other day (Monday, Wednesday, Friday or Tuesday, Thursday, Saturday) across the study period to record menstrual status and collect a 5 ml saliva sample, later assayed for progesterone. Mean peak luteal P4 is defined as the mean daily concentration during the span of days ±2.5 days from the day of observed peak luteal progesterone (i.e. (area under the curve of P4 from x to y) / (y - x) where x to y is any span of days and P4 at any time is defined by linear interpolation of the observed hormone data) [35]. This calculation is typically a more accurate estimate of the highest concentration of luteal progesterone than methods that assume progesterone peaks on a pre-specified cycle day (e.g. 7 days before menstrual bleeding begins). Such an assumption may miss the true peak and underestimate mean peak concentrations due to the considerable inter-cycle variation in the timing of the luteal progesterone peak. As our progesterone variable was positively skewed, the log of progesterone concentration was used in our regression models to avoid violating model assumptions. Only progesterone concentrations in ovulatory cycles were included in our analyses of progesterone levels (anovulatory cycles lack the corpus luteum that produces progesterone; inclusion of anovulatory cycles biases analyses. Supplementary Materials 2 for further justification).

Season was represented by a binary variable of arduous and better seasons across the year. For many women in this population, there are large seasonal differences in food availability, workload, and pathogen exposure. Numbering the days of the year from 1 to 366 (1996 was a leap year), the most arduous seasons consist of harvesting (late rain, early harvest days 1-91) and planting (late-winter/ early-spring, before the summer rains begin, days 245-282) periods, which are more physically demanding than better periods in winter (late harvest, days 92-136, and after harvest before planting begins, days 137-244) and summer (the rainy season when crops are growing, days 283–366). In arduous periods, food availability is relatively lower, labour demands are higher, and pathogen exposure increases. These seasonal timings are based on weather data and direct observation during the year of data collection [36]. The distinction between arduous and better seasons is relative; even during better seasons, life is demanding in these altiplano communities. Arduous seasons account for a shorter proportion of the year at 129 days total, and better seasons account for 237 days total.

Traditional economic score is a measure of each household's subsistence strategy based on principal component analysis of several ethnographically salient socio-economic variables [37]. This factor loads heavily on the husband's frequency of salaried labour (negatively) and on the frequency of agricultural sales by the husband (positively). This factor represents the widespread Andean economic strategy of obtaining income through agricultural sales. Higher household scores reflect sufficient excess agricultural production and sales in the household rather than having to rely on wage labour for income.

Age in years was included to account for proximity to menopause or menarche.

Body fat score is a factor score from a factor analysis on all anthropometric body measurements taken in a given cycle. The factor loaded heavily on all weight, skinfold, and circumference measures, but not height, and explained 58% of the variation in all anthropometrics. This allowed us to capture energetic reserves available to women at the time of haemoglobin measurement.

Breastfeeding status is a binary variable, included to account for the suppressive effects of breastfeeding on ovulation and progesterone. All women in our sample were menstruating, so those who were breastfeeding were not experiencing lactational amenorrhoea.

(g) Statistical analysis

Descriptive analysis presents variation in variables of interest across better seasons and more arduous seasons. To investigate the relationships between socio-ecological factors, haemoglobin, ovulation, and mean peak luteal progesterone, analyses were conducted using multilevel regression models with random intercepts by ID to account for the nonindependence of multiple cycles sampled from the same individual (n = 48 with one cycle, n = 48 with multiple cycles) in this dataset. This was carried out using the glmmTMB pack-(see http://cran.nexr.com/web/packages/glmmTMB/ age glmmTMB.pdf) in R. All continuous independent or adjustment variables were centred on the mean and scaled by dividing by the standard deviation before use in models to obtain comparable effect estimates. Residuals for each model were checked using the DHARMa residual diagnostics package for hierarchical models (https://cran.r-project.org/web/packages/DHARMa/vignettes/DHARMa.html). Models were run (i) for all cycles and (ii) separately for cycles taken in more arduous and better seasons to examine whether the relationships between our variables of interest may be obscured by heterogeneity in seasonal conditions.

3. RESULTS

(a) Descriptive results

The median haemoglobin value was 15.6 g/dl with a range from 9.3 to 20.1 g/dl. Median haemoglobin was 15.15 g/dl in arduous seasons and 15.70 g/dl in better seasons (Table 1). When adjusting for altitude [38], over 25% of the sample was classified as

anaemic (14% with mild anaemia, 8% with moderate anaemia and 3% with severe anaemia) (Fig. 2). In our sample, 71% of cycles in arduous seasons and 78% of cycles in better seasons were ovulatory, and there was a wide range of mean peak luteal progesterone concentration, with a median of 137 pmol/l and an interquartile range of 102–211 pmol/l in arduous seasons and a median of 182 pmol/l and an interquartile range of 130–260 pmol/l in better seasons.

(b) Are haemoglobin levels modulated by the socio-ecology?

To investigate the relationship between socio-ecology and haemoglobin levels we used a linear mixed-effect model to account for repeated cycles within women. We found that on average haemoglobin levels were lower in the arduous seasons than in the better seasons (unadjusted β [95% CI] = 0.76 [0.24, 1.28], P = 0.005) with a predicted average of 14.85 g/dl in the arduous seasons compared to 15.61 g/dl in the better seasons. Evidence for this association remained when adjusting for age (adjusted β [95% CI] = 0.74 [0.22, 1.27], P = 0.006). Further, we found that, in the more arduous seasons, a higher traditional economic score was associated with reduced levels of haemoglobin both independently and after adjusting for age (both P = 0.032) (Table 2, Fig. 3), suggesting that women living in households reliant on traditional agropastoralism had lower levels of haemoglobin in the more arduous seasons compared to those reliant on other forms of income, such as salaried labour. We found no

Table 1. Descriptive statistics for all variables included in the analyses across seasons (N = 152 cycles total)

Variable	N cycles	Arduous seasons, N = 34ª	Better seasons, N = 118ª
Haemoglobin (g/dl)	152	15.15 (13.95, 15.97)	15.70 (14.90, 16.50)
Cycle type	152		
Anovulatory	36	10 (29%)	26 (22%)
Ovulatory	116	24 (71%)	92 (78%)
Mean peak luteal progesterone (pmol/l)	152	137 (102, 211)	182 (130, 260)
Traditional economic score	128	-0.22 (-1.15, 0.85)	-0.15 (-0.70, 0.57)
Age	152	31 (27, 34)	30 (24, 33)
Breastfeeding status	152		
Breastfeeding and menstruating	73	12 (35%)	61 (52%)
Menstruating	79	22 (65%)	57 (48%)
Time since last birth (days)	140	946 (546, 1,857)	612 (434, 1,112)
Body fat score	151	-0.11 (-1.24, 1.65)	0.92 (-1.16, 1.96)

^aMedian (IQR); n (%).



Figure 2. Distribution of haemoglobin values across seasons and anaemia status. The anaemia cut off here is 14.6 g/dl in keeping with the WHO recommendations to raise the anaemia threshold based on residence above sea level [38]

relationship between traditional economic score and haemoglobin in the better seasons.

(c) Is fecundity modulated by haemoglobin levels?

(i) Ovulation

To investigate the relationship between haemoglobin levels and the risk of ovulation, we used a mixed-effect logistic regression model. We found evidence that a one standard deviation increase in haemoglobin levels was associated with around three times the odds of ovulating in a given cycle during the more arduous seasons, both independently (OR = 2.59, 95% CI: [1.03, 6.54], P = 0.043) and when adjusting for age, body fat, and breastfeeding status (OR = 3.27, 95% CI [1.10, 9.77], P = 0.033). We found no relationship between ovulation and haemoglobin levels when including all cycles or cycles occurring during the better seasons (Table 3, Fig. 4).

(ii) Mean peak luteal progesterone

To investigate the relationship between haemoglobin and mean peak luteal progesterone, we used a mixed-effect linear model. Mean peak luteal progesterone was not associated with haemoglobin in ovulatory cycles (Table 4). When adjusting for age, body fat, and breastfeeding status, there was a marginal weak positive relationship between progesterone and haemoglobin in the better seasons with an effect size of 0.08 (P = 0.054), but otherwise there was little evidence to suggest any relationship between haemoglobin and reproductive investment acting through progesterone levels.

4. **DISCUSSION**

This study aimed to investigate the associations between iron status, as measured by haemoglobin levels, and markers of fecundity in a non-contracepting high fertility population from the Bolivian altiplano. We found that (i) haemoglobin levels were lower in arduous compared to better seasons; (ii) in arduous seasons, women in households with higher reliance on agropastoral subsistence strategies had lower haemoglobin levels compared to women in households more reliant on income from salaried labour; (iii) menstrual cycles with lower haemoglobin levels were more likely to be anovulatory during arduous seasons, independently of body fat; and (iv) among ovulatory cycles, haemoglobin levels were not associated with mean peak luteal progesterone. The results have implications for understanding how iron influences the regulation of natural fluctuations in reproductive capabilities.

The connection between haemoglobin levels, season, and traditional modes of subsistence could be influenced by several mechanisms including diminished iron stores (iron-deficiency anaemia) and/or anaemia caused by inflammation. For women reliant on agropastoralism, arduous seasons are characterized by high physical workloads, mostly done with hand tools, and typically involving travelling long distances to the fields for planting and harvesting. There is also reduced food availability and reduced availability of funds to purchase food elsewhere. Most of the time, meat is consumed only rarely and in small portions as part of a soup comprising potatoes, rice, a few vegetables, and occasionally quinoa. Dietary intake of iron is very low, particularly



Table 2. Model output showing crude and age-adjusted estimates for the effect of z-standardized traditional economic scores on haemoglobin levels across seasons

Effect of traditional economic score on haemoglobin	Crude estimates			Age-adjusted estimates			
Sample	Estimate [95% CI]	Standard error	P value	Estimate [95% CI]	Standard error	P value	
All cycles (N = 152)	-0.16 [-0.42, 0.10]	0.13	0.235	-0.17 [-0.43, 0.09]	0.13	0.193	
Better season ($N = 118$)	-0.02 [-0.33, 0.29]	0.16	0.881	-0.04 [-0.35, 0.27]	0.16	0.810	
Arduous season (N = 34)	-0.42 [-0.80, -0.04]	0.20	0.032	-0.42 [-0.81, -0.04]	0.20	0.032	

Bold values indicate P < 0.05.



Figure 3. Predicted values for the age-adjusted relationship between traditional economic score and haemoglobin levels (g/dl) across seasons. Around half of the cycles are the only cycle for that woman and half of the women have multiple cycles. Shaded areas show 95% CIs

in the arduous seasons, and thus, the low haemoglobin levels observed during these periods may indeed represent depleted iron stores [39].

Even if iron stores are sufficient to avoid negative birth outcomes, the body may restrict how much bioavailable iron is circulating in the form of haemoglobin during arduous seasons to mitigate infection [40] when exposure is highest and suppress ovulation to avoid conception during times of low bioavailable iron for pregnancy, favouring investment in immunity over reproductive function. Pathogens need iron to survive and reproduce [41]. For instance, iron is crucial to the life cycle of tuberculosis mycobacteria [42] and an increase in dietary iron has been associated with increased chance of having active pulmonary tuberculosis [43]. In the population under study, exposure to infections is increased during arduous seasons with relatively isolated homesteads coming together for planting and harvesting with increased transmission of any pre-existing infections, weakened immune systems due to taxing workloads, and heighted exposure to infections in the environment, such as waterborne parasites. Risk of infection with tuberculosis and human fascioliasis increases during these seasons [44–47], potentially leading to inflammation-driven reduced iron absorption and reduced release of iron stores, and subsequently lower haemoglobin levels. In this context, restricting the level of bioavailable iron via the hormone hepcidin [2] can be adaptive as it reduces an infection's ability to thrive. To pick apart the role of immune activity in determining haemoglobin levels, future studies should measure both an individual's iron stores, through measuring serum ferritin or

Table 3. Crude and adjusted odds ratios for ovulation with a 1 standard deviation increase in haemoglobin

Effect of haemoglobin on ovulation	Crude odds ratios			Adjusted odds ratios ^a			
Sample	OR [95% CI]	Estimate standard error	P value	OR [95% CI]	Estimate standard error	P value	
All cycles ($N = 152$)	1.40 [0.96, 2.03]	0.19	0.078	1.42 [0.97, 2.09]	0.20	0.074	
Better season ($N = 118$)	1.17 [0.74, 1.84]	0.23	0.499	1.14 [0.72, 1.80]	0.24	0.588	
Arduous season ($N = 34$)	2.59 [1.03, 6.54]	0.47	0.043	3.27 [1.10, 9.77]	0.56	0.033	

Bold values indicate P < 0.05.

^aAdjusted for age, body fat, and breastfeeding status.



Figure 4. Predicted probabilities of ovulation across haemoglobin levels (z standardized) for cycles during the better and more arduous seasons and adjusted for age and breastfeeding status. Shaded areas show 95% CIs

Table 4. Crude and adjusted effect estimates for log-transformed mean peak luteal progesterone with a 1 standard deviation increase in haemoglobin in ovulatory cycles

Effect of haemoglobin on progesterone	Crude estimates			Adjusted estimates ^a			
Sample	Estimate [95% CI]	Standard error	P value	Estimate [95% CI]	Standard error	P value	
All cycles ($N = 152$)	0.05 [-0.03, 0.12]	0.04	0.223	0.06 [-0.01, 0.13]	0.04	0.111	
Better season ($N = 118$)	0.06 [-0.01, 0.14]	0.04	0.103	0.08 [0.00, 0.15]	0.04	0.054	
Arduous season ($N = 34$)	-0.14 [-0.33, 0.05]	0.10	0.144	-0.12 [-0.29, 0.04]	0.09	0.146	

^aAdjusted for age, body fat, and breastfeeding status.

soluble transferrin receptor (sTfR), and markers of acute phase inflammation, such as C-reactive protein (CRP), for detecting the presence of infections.

Our results suggest that haemoglobin levels play a role in the modulation of fecundity, and this effect is independent of energetic stores. Whether driven by reduced iron stores or simply reduced iron availability, in the arduous seasons the risk of ovulation is reduced in the face of low haemoglobin levels. Whilst it might be assumed that women with higher haemoglobin levels may also have improved access to higher quality diets and thus greater energetic availability, instead we find that when adjusting for body fat score, a proxy for energetic status, an independent association between haemoglobin and ovulation remains in the arduous seasons. Both body fat and haemoglobin were positively and independently related to ovulation and as such, each can be considered necessary, but not sufficient, for ovulation to occur. Our findings support calls to focus on resources beyond energy for understanding mechanisms underlying evolutionary and physiological trade-offs [18, 48].

Our findings on the positive association between haemoglobin levels and ovulation risk in arduous but not in better seasons support the idea that organisms have evolved the capability to strategically allocate resources toward present or future reproduction based on the prevailing local socio-ecological conditions. According to the Flexible Response Model of female reproductive function [49, 50], natural selection favours organisms who are able to predict the best conditions they are likely to face in a given environment and, in such conditions, invest in reproduction rather than forego ovulation, even if said conditions are somewhat poor in absolute terms. Reduced investment in current reproduction may help avoid negative pregnancy outcomes [6] until iron stores are repleted, such as in better seasons when haemoglobin levels improve for women relying on agropastoralism. However, if haemoglobin levels are still low in the better seasons, there is reduced benefit to delaying reproduction further given that conditions are unlikely to improve. We include an exploration of whether women in our sample with low haemoglobin predictively suppressed ovulation at times where haemoglobin would subsequently increase in Supplementary Materials 3. This analysis is not included here as it is substantially limited by low power, but the direction of effect is supportive of women flexibly delaying ovulation when haemoglobin is low but will improve, and provides a springboard for future research.

Unlike for ovulation, we find little evidence to suggest that haemoglobin is associated with variation in mean peak luteal progesterone levels in ovulatory cycles. Firstly, this may be due to the index of progesterone we used (mean peak luteal progesterone), which reflects the absolute level of progesterone on and about the peak production of this hormone but does not reflect the magnitude of change from levels immediately after ovulation to the levels occurring at peak production. Recent research has suggested that *change* may actually be more biologically salient than *level* itself [18]. Secondly, we may fail to see a relationship due to sampling error as progesterone release is triggered in a pulsatile manner with wide variation in levels observed across the day depending on proximity to the pulsatile signal [51]. Finally, it could be that haemoglobin is not very relevant to the modulation of progesterone levels and the subsequent decidualisation of the endometrium and/or that progesterone levels above a necessary minimum are not essential to successful continuation of pregnancy during the weeks before the shift from luteal to predominantly placental production of progesterone [22, 26].

The main strength of our analysis lies in its ability to reveal human life history trade-offs that would otherwise not manifest or be statistically detectable, through the specificities of lives of Aymara women living at high-altitude in the Bolivian altiplano. Trade-offs are sometimes obscured by heterogeneity across a population or across time in access to resources or conditions experienced (Bolund, 2020). Our ability to analyse cycles collected from different seasons of the year among women utilizing different subsistence strategies allows us to investigate how trade-offs change over time and over conditions with the use of multilevel modelling to consider repeat measurements within the same women. We were limited in our ability to fully understand the kind of trade-offs we are observing and the mechanisms which underpin them without access to markers of infection, immune function, and iron stores. However, to collect extra markers in a large sample would have represented a significant extra cost and toll on participants, and highlights how studying the mechanisms underpinning trade-offs in humans is ethically and practically demanding [52]. Additionally, there are several potential selection biases which may have influenced our results. For instance, as we only included non-conceptive cycles, we have an overrepresentation of 'non-successful' cycles and so we cannot make inferences about the relationship between haemoglobin and ovulation in cycles that led to pregnancy.

Currently, it is recommended that iron supplements, typically in combination with folic acid, are given to general populations of women of reproductive age and pregnant women across the globe [53]. However, uncritical provision of supplementation without considering the trade-offs individuals face from a life history perspective has the potential to cause harm [54]. For instance, prenatal protein-energy supplementation has been shown to reduce subsequent birth intervals to the next child [55], with tightly spaced pregnancies taking a physical toll on women and, in some cases, returning the mother to a state of undernourishment. Given our findings, it is possible that iron supplementation of iron-deficient postpartum women, with the intention to raise iron stores and subsequent haemoglobin levels, could reduce the time to return of ovulation and shorten birth intervals with similar detrimental effects to protein-energy supplementation, particularly if the individual has been unable to replete other bodily reserves in time or is still breastfeeding. Additionally, if iron supplements are given in cases of non-iron-deficiency anaemia, most commonly driven by infection [56], this risks either being, at best, largely ineffective as iron gut absorption is heavily limited by hepcidin, or, at worst, puts women at a greater risk of illness if an infection is able to utilize the extra iron to proliferate [40]. Thus, it is crucial to contemplate iron supplementation only after evaluating a woman's broader immune and reproductive circumstances and determining the specific type of anaemia she is experiencing.

In conclusion, our results among Aymara women living in the Bolivian altiplano support the hypothesis that fecundity is modulated based on iron status beyond energy intake. The potential for ovulation to be inhibited due to low haemoglobin is demonstrated, and this correlation exhibits a flexible response to the existing socio-economic factors and whether there is potential for seasonal improvement in conditions. Thus, we propose that iron status may be modulating resource allocation trade-offs between current and future reproduction and/or between immunity and reproduction and that these trade-offs should be considered when designing interventions to reduce anaemia.

SUPPLEMENTARY DATA

Supplementary data is available at EMPH online.

AUTHOR ROLES

RS contributed to the inception of the research question and conducted the data analysis and led on writing the manuscript. AA contributed to the inception of the research question, supervised the data analysis and writing of the manuscript by RS and provided input at all stages and guidance on analytical decisions. VJV collected and provided the data, contributed to the inception of the research question, contributed information for sections on project background and data collection methods, and provided input on analyses and manuscript copy at all stages.

AUTHOR CONTRIBUTIONS

Rose Stevens (Conceptualization [Lead], Formal analysis [Lead], Methodology [Equal], Visualization [Lead], Writing—original draft [Lead], Writing—review & editing [Lead]), Aexandra Alvergne (Conceptualization [Supporting], Formal analysis [Supporting], Investigation [Supporting], Methodology [Equal], Supervision [Equal], Writing—review & editing [Supporting]) and Virginia Vitzthum (Conceptualization [Supporting], Data curation [Lead], Funding acquisition [Lead], Supervision [Equal], Writing—review & editing [Supporting])

ACKNOWLEDGEMENTS

We thank the Economic and Social Research Council (ESRC, ES/P000649/1) and the US National Science Foundation (SBR 9506107, SBR 9507320) for funding this work; the Instituto Boliviano de Biología de Altura (IBBA) for facilities and logistical support; Lic. Esperanza Caceres for her invaluable contributions throughout data collection, Dr Jonathan Thornburg for analytical input, and the Aymara women for their time and participation.

This work is dedicated to the memory of Dr. Hilde Spielvogel, core investigator with Project REPA and Director of the Department of Bioenergetics (IBBA), Facultad de Medicina, Universidad Mayor de San Andrés, La Paz, Bolivia. Her keen intellect, good humour, generosity and kindness to innumerable patients, students, colleagues and collaborators enriched all and immeasurably advanced understanding of adaptations and health in highaltitude populations.

STUDY FUNDING

Funded by the Economic and Social Research Council (ESRC, ES/ P000649/1) and the US National Science Foundation (SBR 9506107, SBR 9507320).

CONFLICT OF INTEREST

None declared.

DATA AVAILABILITY

The data underlying this analysis is available via Dryad (access DOI: 10.5061/dryad.tqjq2bw7f).

REFERENCES

- Safiri S, Kolahi AA, Noori M *et al*. Burden of anemia and its underlying causes in 204 countries and territories, 1990–2019: results from the Global Burden of Disease Study 2019. *J Hematol Oncol J Hematol Oncol* 2021;**14**:185.
- 2. Chaparro CM, Suchdev PS. Anemia epidemiology, pathophysiology, and etiology in low- and middle-income countries. *Ann N Y Acad Sci* 2019;**1450**:15–31.
- Bedwell RM, Spielvogel H, Bellido D *et al*. Factors influencing the use of biomedical health care by rural Bolivian anemic women: structural barriers, reproductive status, gender roles, and concepts of anemia. *PLoS One* 2017;12:e0170475–20.
- Vitzthum VJ. The ecology and evolutionary endocrinology of reproduction in the human female. Am J Phys Anthropol 2009;140:95–136.
- 5. Wade J, Byrne DJ, Ballentine CJ *et al.* Temporal variation of planetary iron as a driver of evolution. *Proc Natl Acad Sci USA* 2021;**118**:e2109865118.
- Scholl TO. Iron status during pregnancy: setting the stage for mother and infant. Am J Clin Nutr 2005;81:1390–6. https://pubmed.ncbi.nlm. nih.gov/15883455/.

- Miller EM. The reproductive ecology of iron in women. Am J Phys Anthropol 2016;159:S172–95.
- Sear R. Evolutionary Demography of a Rural Gambian Population. 2001, 247. University College London, 2001. ProQuest LLC(2016).
- Sear R, Mace R, McGregor IA. A life history approach to fertility rates in rural gambia: evidence for trade-offs or phenotypic correlations? In: Rodgers JL, Kohler HP (eds.). *The Biodemography of Human Reproduction and Fertility*. Boston, MA: Springer US, 2003, 135–60. doi:10.1007/978-1-4615-1137-3_7
- Clancy KBH, Nenko I, Jasienska G. Menstruation does not cause anemia: endometrial thickness correlates positively with erythrocyte count and hemoglobin concentration in premenopausal women. *Am J Hum Biol* 2006;**18**:710–3.
- Miller EM, Khalil M. Iron and fecundity among Tsimane' women of Bolivia. Evol. Med. Public Health 2019;2019:111-20.
- 12. Roff DA. The Evolution of Life Histories: Theory and Analysis. NY: Chapman & Hall, 1992.
- 13. Stearns SC. The Evolution of Life Histories. Oxford: Oxford University Press, 1992.
- 14. Prior JC. Luteal phase defects and anovulation: adaptive alterations occurring with conditioning exercise. *Semin Reprod Med* 1985;3:27-33.
- Prior JC. Physical exercise and the neuroendocrine control of reproduction. Baillieres Clin Endocrinol Metab 1987;1:299–317.
- Vitzthum VJ. Evolutionary models of women's reproductive functioning. Annu Rev Anthropol 2008;37:53–73.
- 17. Ellison PT. Reproductive Ecology and Human Evolution. 1st edn. NY: Routledge, 2011.
- Vitzthum VJ. How it works: the biological mechanisms that generate demographic diversity. In: Burger O, Lee R, Sear R (eds.). *Human Evolutionary Demography*. Cambridge: Open Book Publishers, 2024, 251–90. doi: 10.11647/OBP.0251
- Miller EM. Maternal hemoglobin depletion in a settled Northern Kenyan pastoral population. Am J Hum Biol 2010;22:768-74.
- Aleshire S, Osteen K, Maxson W *et al.* Localization of transferrin and its receptor in ovarian follicular cells: morphologic studies in relation to follicular development. *Fertil Steril* 1989;51:444–9.
- Kolesarova A, Capcarova M, Medvedova M *et al*. In vitro assessment of iron effect on porcine ovarian granulosa cells: secretory activity, markers of proliferation and apoptosis. *Physiol Res* 2011;**60**:503–10.
- Palomba S, Santagni S, La Sala GB. Progesterone administration for luteal phase deficiency in human reproduction: an old or new issue? J Ovarian Res 2015;8:1–15.
- 23. Beall CM, Brittenham GM, Strohl KP *et al*. Hemoglobin concentration of high-altitude Tibetans and Bolivian Aymara. *Am J Phys Anthropol* 1998;**106**:385–400.
- 24. Beall CM. Andean, Tibetan, and Ethiopian patterns of adaptation to high-altitude hypoxia. *Integr Comp Biol* 2006;**46**:18–24.
- Gassmann M, Mairbäurl H, Livshits L *et al*. The increase in hemoglobin concentration with altitude varies among human populations. *Ann N Y Acad Sci* 2019;**1450**:204–20.
- Vitzthum VJ, Spielvogel H, Thornburg J. Interpopulational differences in progesterone levels during conception and implantation in humans. *Proc Natl Acad Sci U S A* 2004;**101**:1443–8.
- Angeli A, Lainé F, Lavenu A *et al.* Joint model of iron and hepcidin during the menstrual cycle in healthy women. AAPS J 2016;18:490–504.

- Sterne JAC, White IR, Carlin JB *et al*. Multiple imputation for missing data in epidemiological and clinical research: potential and pitfalls. *BMJ* 2009;**338**:b2393.
- Mayer M. missRanger: Fast Imputation of Missing Values. 2017.
 p. 2.4.0. https://CRAN.R-project.org/package=missRanger (1 June 2024, date last accessed).
- Van Buuren S, Groothuis-Oudshoorn K. mice: Multivariate Imputation by Chained Equations. 2006. p. 3.16.0. https://CRAN.R-project.org/package=mice (1 June 2024, date last accessed).
- Hernán MA. The C-word: scientific euphemisms do not improve causal inference from observational data. Am J Public Health 2018;108:616–9.
- Rohrer JM. Thinking clearly about correlations and causation: graphical causal models for observational data. *Adv Meth Pract Psychol Sci* 2018;1:27–42.
- Tennant PWG, Murray EJ, Arnold KF *et al.* Use of directed acyclic graphs (DAGs) to identify confounders in applied health research: review and recommendations. *Int J Epidemiol* 2021;**50**:620–32.
- Prabhakaran S. r-statistics.co. 2016. Outlier Treatment With R. http://r-statistics.co/Outlier-Treatment-With-R.html (1 June 2024, date last accessed).
- Vitzthum VJ. Field methods and strategies for assessing female reproductive functioning. *Am J Hum Biol* 2021;33:e23513.
- Vitzthum VJ, Thornburg J, Spielvogel H. Seasonal modulation of reproductive effort during early pregnancy in humans. *Am J Hum Biol* 2009;21:548–58.
- Bindon JR, Vitzthum VJ. Household economic strategies and nutritional anthropometry of women in American Samoa and highland Bolivia. *Soc Sci Med* 2002;54:1299–308.
- 38. World Health Organization. Guideline on haemoglobin cutoffs to define anaemia in individuals and populations. Geneva, 2024. https://iris.who. int/bitstream/handle/10665/376196/9789240088542-eng.pdf?sequence=1 (1 June 2024, date last accessed).
- Berti PR, Fallu C, Agudo YC. A systematic review of the nutritional adequacy of the diet in the Central Andes. *Rev Panam Salud Publica* 2014;10:314–23.
- Pasricha SR, Armitage AE, Prentice AM *et al.* Reducing anaemia in low income countries: control of infection is essential. *BMJ* 2018;362:k3165.
- Drakesmith H, Prentice AM. Hepcidin and the Iron-Infection Axis. Science 2012;338:768–72. DOI: 10.1126/science.122457.
- Tanner R, O'Shea MK, White AD *et al*. The influence of haemoglobin and iron on in vitro mycobacterial growth inhibition assays. *Sci Rep* 2017;**7**:1–13.
- Gangaidzo IT, Moyo VM, Mvundura E et al. Association of pulmonary tuberculosis with increased dietary iron. J Infect Dis 2001;184:936–9.
- 44. Fares A. Seasonality of tuberculosis. J Glob Infect Dis 2011;3:46–55.
- 45. World Health Organization. Groups at risk: WHO report on the tuberculosis epidemic 1996. Geneva: World Health Organization, 1996. https://scholar.google.com/scholar_lookup?title=Report%20on%20the%20 Tuberculosis%20Epidemic%201997&publication_year=1998#d=gs_ cit&t=1661254823579&u=%2Fscholar%3Fq%3Dinfo%3A8tUX00rUC-MgJ%3Ascholar.google.com%2F%26output%3Dcite%26scirp% 3D0%26hl%3Den
- Parkinson M, O'Neill SM, Dalton JP. Endemic human fasciolosis in the Bolivian Altiplano. *Epidemiol Infect* 2007;135:669–74.

- Caravedo MA, Cabada MM. Human fascioliasis: current epidemiological status and strategies for diagnosis, treatment, and control. *Res Rep Trop Med* 2020;11:149–58.
- Flatt T, Heyland A. Mechanisms of life history evolution: the genetics and physiology of life history traits and trade-offs. *Mechanisms of Life History Evolution*. Oxford: Oxford University Press, 2013.
- Vitzthum VJ. Why not so great is still good enough flexible responsiveness in human reproductive functioning. In: Ellison PT (ed.). *Reproductive Ecology and Human Evolution*. 1st ed. NY: Aldine Transaction, 2001, 179–202.
- 50. Vitzthum VJ, Spielvogel H. Epidemiological transitions, reproductive health, and the flexible response model. *Econ Hum Biol* 2003;1:223–42.
- Filicori M, Butler JP, Crowley WF. Neuroendocrine regulation of the corpus luteum in the human. Evidence for pulsatile progesterone secretion. J Clin Invest 1984;73:1638–47.

- 52. Bolund E. The challenge of measuring trade-offs in human life history research. *Evol Hum Behav* 2020;**41**:502–12.
- World Health Organization. Global Anaemia Reduction Efforts Among Women of Reproductive Age: Impact, Achievement of Targets and the Way Forward for Optimizing Efforts. Geneva: World Health Organization, 2020.
- Reiches MWA. life history approach to prenatal supplementation: Building a bridge from biological anthropology to public health and nutrition. *Am J Hum Biol* 2019;**31**:e23318.
- Lunn PG, Austin S, Prentice AM *et al*. The effect of improved nutrition on plasma prolactin concentrations and postpartum infertility in lactating Gambian women. *Am J Clin Nutr* 1984;**39**:227–35.
- Petry N, Olofin I, Hurrell RF *et al.* The proportion of anemia associated with iron deficiency in low, medium, and high human development index countries: a systematic analysis of national surveys. *Nutrients* 2016;8:693.